Author's response to reviews

Title: Attitudes among healthcare professionals to the reporting of adverse drug reactions in Nepal

Authors:

Santosh KC (sant_02@hotmail.com)
Pramote Tragulpiankit (pramote.tra@mahidol.ac.th)
Sarun Gorsanan (sarun103@yahoo.com)
Ralph Edwards I (ralph.edwards@who-umc.org)

Version: 2 Date: 9 August 2012

Author's response to reviews: see over
To

The Editor- in- Chief

BMC Clinical Pharmacology

Subject: Submission of manuscript

Dear Sir,

Herewith, we are submitting an article titled “Attitudes among healthcare professionals to the reporting of adverse drug reactions reporting in Nepal” for publication in your esteemed journal. All of the authors listed have gone through the manuscript and approved the same for publication. We hope this article would be of great help to contribute to the knowledge on pharmacovigilance in the resource limiting settings. This is the first article related to attitudes towards adverse drug reactions reporting in Nepal among healthcare professionals. This is an original research article and we have not submitted to any other journal for the consideration. We do not have any financial interests or other conflicts either directly or indirectly associated with this article. We have taken all the efforts to comply with the manuscript checklist of the journal. Kindly consider our article for publication.

Thanking you in anticipation.

Yours sincerely

Pramote Tragulpiankit
Department of Pharmacy, Faculty of Pharmacy, Mahidol University,
447 Sri-Ayudhaya Road, Bangkok 10400, Thailand
Tel: (662) 6448694
Fax: (662) 6448694
E-mail: pyptg@mahidol.ac.th